Author's response to reviews

Title: Testing use of payers to facilitate evidence-based practice adoption. Protocol for a cluster-randomized trial.

Authors:

Todd D Molfenter (todd.molfenter@chess.wisc.edu)
Jee-Seon Kim (jeeseonkim@wisc.edu)
Andrew Quanbeck (andrew.quanbeck@chess.wisc.edu)
Terry Patel-Porter (Terry.Patel@ada.ohio.gov)
Sandy Starr (SStarr@ada.ohio.gov)
Dennis McCarty (mccartyd@ohsu.edu)

Version: 3 Date: 29 April 2013

Author's response to reviews:

Dear Editor(s):

Attached are the final proofs for Molfenter et. al. manuscript. Two manuscript files will be uploaded. The edited version in the manuscript location and a version of the manuscript with track changes visible (this version also has the editors comments included).

All changes made by the editor were accepted. In the manuscript, two sections were edited by the authors based on editor suggestions: one section in the discussion section of the abstract and the other section in the Data Analysis on Page 12.

Thank you for your guidance with the manuscript and please let me know if there are any further questions.

Sincerely, Todd Molfenter